Title: Intestinal neuroendocrine tumor in a patient with pituitary adenoma. Case report and review of the current screening recommendations.

Authors:

Cherif Boutros (cherifboutros@hotmail.com)  
Diana Cheng-Robles (chengrdi@umdnj.edu)  
Robert Goldenkranz (rgoldenkranz@sbhcs.com)

Version: 2 Date: 30 June 2007

Author's response to reviews:

Please insert an Abstract onto page 2.  
- Done

Please insert a Competing interests section after the Conclusions. If the authors have no competing interests, please state: "The authors declare that they have no competing interests."
- Done

Please place the Authors' contributions section after the Conclusions.  
- Done

Please detail whether written informed patient consent was obtained for publication within an Acknowledgements section.  
- Done

It is journal style to list no more than 15 references - please cut down the reference list to between 10 and 15 references.  
- Done